Author's response to reviews

Title: The Incidence of Adverse Events in an Italian Acute Care Hospital. Findings of a Two-stage Method in a retrospective cohort study.

Authors:

Lorenzo Sommella (l.sommella@sanfilipponeri.roma.it)
Chiara de Waure (chiara.dewaure@rm.unicatt.it)
Anna M Ferriero (annamariaferriero@hotmail.it)
Amalia Biasco (biasco.amalia@gmail.com)
Maria T Mainelli (mariateresa_mainelli@yahoo.it)
Luigi Pinnarelli (luigipinnarelli@yahoo.it)
Walter Ricciardi (wricciardi@rm.unicatt.it)
Gianfranco Damiani (gdamiani@rm.unicatt.it)

Version: 2
Date: 24 February 2014

Author's response to reviews: see over
Rome, February, 24th 2014

Prof. G Damiani
Department of Public Health
Catholic University of Sacred Heart,
L.go Francesco Vito 1, 00168 Rome, Italy

Subject: Manuscript Submission
Direct line: +39 06 30153837
Fax number: +39 06 35001522
E-mail: gdamiani@rm.unicatt.it

Dear Editor,

We are writing you regarding the paper “The Incidence of Adverse Events in an Italian Acute Care Hospital. Findings of a Two-stage Method in a retrospective cohort study”, submitted to BMC Health Services Research.

As requested, we amended the manuscript by adding:

- Abstract,
- Ethics Statement,
- Authors’ Contributions.

If you need further information, please do not hesitate to contact us.

Best regards,

Lorenzo Sommella
Chiara de Waure
Anna Maria Ferriero
Amalia Biasco
Maria Teresa Mainelli
Luigi Pinnarelli
Walter Ricciardi
Gianfranco Damiani